Author's response to reviews

Title: Evaluation of community level interventions to address social and structural determinants of health: a cluster randomised controlled trial.

Authors:

Martin Wall (m.wall@massey.ac.nz)  
Richard Hayes (richard.hayes@lshtm.ac.uk)  
Derek Moore (d.g.moore@uel.ac.uk)  
Mark Petticrew (mark.petticrew@lshtm.ac.uk)  
Angela Clow (clowa@westminster.ac.uk)  
Elena Schmidt (e.schmidt@uel.ac.uk)  
Alizon Draper (drapera@westminster.ac.uk)  
Karen Lock (karen.lock@lshtm.ac.uk)  
Rebecca Lynch (r.lynnch@ucl.ac.uk)  
Adrian Renton (a.renton@uel.ac.uk)

Version: 2 Date: 18 June 2009

Author's response to reviews:

Cover letter for revised version  
18th June 2009

Dear Editors

Thank you for your email.

Under Major revisions I have made the following changes:

Author List - All the authors' names match those in the submission system. Note I had to change one name 'Elena Schmidt' in the submission system as it was the wrong way round. I have removed job titles.

I have made changes to the references 6, 11, 16, 21 and 58 as requested. These changes have also been made in the accompanying endnote file etl.enl.

Minor Revisions

Author List. I have reformatted as a paragraph and used superscripts to indicate affiliations.

I have included email addresses in the format requested.

I have removed the figure placement text.

I have replaced underlined text with bold except where the underlining indicates a hyperlink or email.
I have put a sub-heading 'references' above the reference list.

I have not further cropped the figure as when I try and do this the quality of the figure deteriorates. One of the other authors originally created this figure and I can request a better version from him if the quality is too poor.

Martin Wall